Author's response to reviews

Title: A case of solid pseudopapillary neoplasm without cysts that occurred in the young man diagnosed by EUS-FNA

Authors:

Masakuni Fujii (sktnq334@yahoo.co.jp)
Masao Yoshioka (m-yoshioka@saiseidr.jp)
Takefumi Niguma (niguma@saiseidr.jp)
Hiroaki Saito (gentetu3110@gmail.com)
Toru Kojima (trkojima0507@yahoo.co.jp)
Soichiro Nose (nos@saiseidr.jp)
Junji Shiode (siodeju@gmail.com)

Version: 2 Date: 2 April 2014

Author's response to reviews: see over
March 29, 2014
Prof. Michael Kidd
Editor-in-Chief,
Journal of Medical Case Reports,
c/o BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Prof. Michael Kidd

Title: A case of solid pseudopapillary neoplasm without cysts that occurred in the young man diagnosed by EUS-FNA

On behalf of all the authors, I would like to ask you to consider our manuscript entitled for publication in Journal of Medical Case Reports as a Case Report. We believe that this is an important aspect and worthy of publication in the Journal. This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and have approved this submission. We have no disclosure and financial support. We report no conflicts of interest.

Sincerely yours,

Corresponding Author
Masakuni Fujii, MD
Department of Internal Medicine, Okayama Saiseikai General Hospital, 1-17-18, Ifuku-cho, Okayama, 700-8511, Japan.
Tel: +81-86-252-2211
E mail address: sktnng334@yahoo.co.jp